Author's response to reviews

Title: Rapid detection of SMARCB1 sequence variation using high resolution melting.

Authors:

Vinod Dagar (vinod.dagar@mcri.edu.au)
Chung-Wo Chow (cw.chow@rch.org.au)
David M Ashley (david.ashley@rch.org.au)
Elizabeth M Algar (elizabeth.algar@rch.org.au)

Version: 3 Date: 3 December 2009

Author's response to reviews:

Dear Editors

I have made the following formatting changes to the manuscript (Version 3)

Major

Author list style has been amended.

Author contributions have been amended to comply with the recommended format.

Table 2 has now been cited in the manuscript.

Minor.

The country has been added to the author affiliations.

Underlines have been removed from the text.

The word count has been removed.

Abbreviations are listed in paragraph format.

Colour text has been changed to black.

Vertical lines have been removed from tables.

Some small typographical errors have been amended.

There have been no changes to the figures.

Yours sincerely

Elizabeth Algar